

# 

**Citation:** Gurol-Urganci I, Waite L, Webster K, Jardine J, Carroll F, Dunn G, et al. (2022) Obstetric interventions and pregnancy outcomes during the COVID-19 pandemic in England: A nationwide cohort study. PLoS Med 19(1): e1003884. https:// doi.org/10.1371/journal.pmed.1003884

Academic Editor: Gordon C. Smith, Cambridge University, UNITED KINGDOM

Received: April 24, 2021

Accepted: December 7, 2021

Published: January 10, 2022

**Copyright:** © 2022 Gurol-Urganci et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results presented in the study can be requested from NHS Digital (https://digital.nhs.uk/services/ data-access-request-service-dars/dars-productsand-services).

**Funding:** The National Maternity and Perinatal Audit (NMPA) is commissioned by the Healthcare Quality Improvement Partnership (HQIP; <u>www.</u> <u>hqip.org.uk</u>) as part of the National Clinical Audit and Patient Outcomes Programme and funded by NHS England and the Scottish and Welsh **RESEARCH ARTICLE** 

# Obstetric interventions and pregnancy outcomes during the COVID-19 pandemic in England: A nationwide cohort study

Ipek Gurol-Urganci<sup>1,2°</sup>, Lara Waite<sup>1°</sup>, Kirstin Webster<sup>1°</sup>, Jennifer Jardine<sup>1,2°</sup>, Fran Carroll<sup>1</sup>, George Dunn<sup>1</sup>, Alissa Frémeaux<sup>1</sup>, Tina Harris<sup>3</sup>, Jane Hawdon<sup>4</sup>, Patrick Muller<sup>1,2°</sup>, Jan van der Meulen<sup>2‡</sup>, Asma Khalil<sup>5,6,7‡</sup>\*

1 Royal College of Obstetricians and Gynaecologists, London, United Kingdom, 2 Department of Health Services Research, Faculty of Public Health and Policy, London School of Hygiene & Tropical Medicine, London, United Kingdom, 3 Faculty of Health and Life Sciences, De Montfort University, Leicester, United Kingdom, 4 Royal Free London NHS Foundation Trust, London, United Kingdom, 5 Fetal Medicine Unit, St George's Hospital, London, United Kingdom, 6 Vascular Biology Research Centre, Molecular and Clinical Sciences Research Institute, St George's, University of London, London, United Kingdom, 7 Fetal Medicine Unit, Liverpool Women's Hospital, Liverpool, United Kingdom

• These authors contributed equally to this work.

‡ These authors are joint senior authors on this work.

\* akhalil@sgul.ac.uk

# Abstract

# Background

The COVID-19 pandemic has disrupted maternity services worldwide and imposed restrictions on societal behaviours. This national study aimed to compare obstetric intervention and pregnancy outcome rates in England during the pandemic and corresponding pre-pandemic calendar periods, and to assess whether differences in these rates varied according to ethnic and socioeconomic background.

# Methods and findings

We conducted a national study of singleton births in English National Health Service hospitals. We compared births during the COVID-19 pandemic period (23 March 2020 to 22 February 2021) with births during the corresponding calendar period 1 year earlier. The Hospital Episode Statistics database provided administrative hospital data about maternal characteristics, obstetric inventions (induction of labour, elective or emergency cesarean section, and instrumental birth), and outcomes (stillbirth, preterm birth, small for gestational age [SGA; birthweight < 10th centile], prolonged maternal length of stay ( $\geq$ 3 days), and maternal 42day readmission). Multi-level logistic regression models were used to compare intervention and outcome rates between the corresponding pre-pandemic and pandemic calendar periods and to test for interactions between pandemic period and ethnic and socioeconomic background. All models were adjusted for maternal characteristics including age, obstetric history, comorbidities, and COVID-19 status at birth. The study included 948,020 singleton births (maternal characteristics: median age 30 years, 41.6% primiparous, 8.3% with gestational diabetes, 2.4% with preeclampsia, and 1.6% with pre-existing diabetes or Governments. All authors are members of the NMPA project team. Neither HQIP nor the funders had a role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** All authors have received support from Healthcare Quality Improvement Partnership. JJ and LW are or have been members of the RCOG COVID-19 guidance cell which produces clinical guidance and policy documents to support the management of pregnant women during the pandemic in the UK. AK is PI of COVID-19 in pregnancy studies and a member of the COVAX Working Group on COVID-19 in pregnancy.

Abbreviations: aOR, adjusted odds ratio; HES, Hospital Episode Statistics; ICD-10, International Classification of Diseases–10th revision; IMD, Index of Multiple Deprivation; NHS, National Health Service; ONS, Office for National Statistics; OPCS-4, Office for Population Censuses and Surveys Classification of Interventions and Procedures–4th revision; OR, odds ratio; SGA, small for gestational age. hypertension); 451,727 births occurred during the defined pandemic period. Maternal characteristics were similar in the pre-pandemic and pandemic periods. Compared to the prepandemic period, stillbirth rates remained similar (0.36% pandemic versus 0.37% pre-pandemic, p = 0.16). Preterm birth and SGA birth rates were slightly lower during the pandemic (6.0% versus 6.1% for preterm births, adjusted odds ratio [aOR] 0.96, 95% CI 0.94–0.97; 5.6% versus 5.8% for SGA births, aOR 0.95, 95% CI 0.93–0.96; both p < 0.001). Slightly higher rates of obstetric intervention were observed during the pandemic (40.4% versus 39.1% for induction of labour, aOR 1.04, 95% CI 1.03–1.05; 13.9% versus 12.9% for elective cesarean section, aOR 1.13, 95% CI 1.11–1.14; 18.4% versus 17.0% for emergency cesarean section, aOR 1.07, 95% CI 1.06–1.08; all p < 0.001). Lower rates of prolonged maternal length of stay (16.7% versus 20.2%, aOR 0.77, 95% CI 0.76–0.78, p < 0.001) and maternal readmission (3.0% versus 3.3%, aOR 0.88, 95% CI 0.86–0.90, p < 0.001) were observed during the pandemic period. There was some evidence that differences in the rates of preterm birth, emergency cesarean section, and unassisted vaginal birth varied according to the mother's ethnic background but not according to her socioeconomic background. A key limitation is that multiple comparisons were made, increasing the chance of false-positive results.

# Conclusions

In this study, we found very small decreases in preterm birth and SGA birth rates and very small increases in induction of labour and elective and emergency cesarean section during the COVID-19 pandemic, with some evidence of a slightly different pattern of results in women from ethnic minority backgrounds. These changes in obstetric intervention rates and pregnancy outcomes may be linked to women's behaviour, environmental exposure, changes in maternity practice, or reduced staffing levels.

# Author summary

# Why was this study done?

- The COVID-19 pandemic has been associated with a reduction in the rates of preterm births and small-for-gestational-age births in some high-income settings.
- It is unknown how changes in high-income settings have influenced rates of obstetric interventions, and whether the pandemic has exacerbated existing inequalities in maternity care and outcomes.

# What did the researchers do and find?

• We conducted a national population-based study, using administrative hospital records from England to compare obstetric intervention rates and maternal outcomes during the pandemic period and the corresponding calendar period 1 year earlier.

- Rates of obstetric interventions were slightly higher in England during the pandemic than in the pre-pandemic period.
- There was some evidence of slightly lower rates of preterm birth and small-for-gestational-age birth.
- There were very small variations in the pattern of results according to women's ethnic background.

#### What do these findings mean?

- The COVID-19 pandemic was not associated with overall worse pregnancy outcomes in England. However, there was a higher rate of obstetric interventions, which may have been the consequence of lockdown restrictions and service interruptions.
- The results should be interpreted with caution. All differences were small, and many comparisons were made.

# Introduction

During the COVID-19 pandemic, pregnant women are vulnerable to both the 'direct effects' of infection with SARS-CoV-2 and the 'indirect effects' of disruption of essential healthcare services and restrictions on social interaction. There is no evidence that pregnant women face an increased risk of contracting SARS-CoV-2 infection compared to non-pregnant women. However, there is evidence that women who experience a SARS-CoV-2 infection during pregnancy may have higher rates of maternal morbidity and mortality including preterm birth, preeclampsia, and emergency cesarean delivery; neonatal morbidity; and perinatal morbidity and mortality including stillbirth [1–4]. The most recent data from the UK Obstetric Surveillance System give a preterm birth rate of 21% and a cesarean birth rate of 43% for pregnant women who require admission to hospital with a SARS-CoV-2 infection [5].

There are reports that the impact of COVID-19 differs among ethnic and socioeconomic groups, with pregnant women from ethnic minority backgrounds more likely to be hospitalised [5]. Furthermore, the direct impact of the SARS-CoV-2 infection has been disproportionately high in ethnic minority groups and socioeconomically deprived communities [6].

The population-level impacts of the indirect effects of the pandemic on maternal and neonatal outcomes are, however, likely to be larger than direct effects of COVID-19 infection [7– 9]. A recent systematic review and meta-analysis focusing on the indirect effects of the pandemic on maternity outcomes found that in low- and middle-income settings, the pandemic period has been associated with an increase in maternal mortality and stillbirths [7]. However, to the best of our knowledge, there is no evidence of such an association in high-income settings, and uncertainty remains about the broader impact of the pandemic.

Almost all available reports, with the exception of national data from the US, are restricted to the first 8 months of 2020 [7,10]. Consequently, there is little evidence regarding the indirect effects of the pandemic during the later months of 2020, when many women would have experienced changes to care and restrictions on social interaction for most or all of their pregnancy.

In this study, we used population-level data from England to examine the combined direct and indirect effects of the COVID-10 pandemic on mode of delivery and pregnancy outcomes. We compared obstetric intervention and pregnancy outcome rates in England during a defined 'pandemic period' (23 March 2020 to 22 February 2021, including 2 periods of national 'lockdown') with those during the corresponding pre-pandemic calendar period 1 year earlier (23 March 2019 to 22 February 2020) and tested whether differences in these rates varied according to ethnic and socioeconomic background.

During the pandemic period, there were major changes to healthcare delivery throughout the year [11]. For maternity care, there has been consistent national provision of rapid guidance and support from professional organisations to inform changes in service [12]. Changes were broadly similar to those seen across Europe, including increased use of remote consultation, reduced provision of out-of-hospital care and birth, and widespread restrictions on access for support partners [13].

#### Methods

#### Study design

This study is a national population-based study using Hospital Episode Statistics (HES) records of all inpatient admissions to National Health Service (NHS) hospitals in England [14]. HES records contain data on patient demographics (age, sex, and ethnicity), admission and discharge dates, and diagnostic and procedure information. Diagnostic information is coded using the International Classification of Diseases–10th revision (ICD-10) [15]. Procedures are coded using the UK Office for Population Censuses and Surveys Classification of Interventions and Procedures–4th revision (OPCS-4) [16]. Each episode related to the birth of a baby contains details of the labour and birth (e.g., gestational age and birthweight) in supplementary data fields known as the HES 'maternity tail'.

#### Cohort selection and outcome definitions

The pandemic period and the lockdown periods were defined according to the timeline of events related to the COVID-19 pandemic in England: the start of the first national lockdown (23 March 2020), the relaxation of lockdown restrictions and move to a tiered system of local restrictions (24 June 2020), the start of national measures leading to a second national lockdown (22 September 2020), and the relaxation of national lockdown restrictions (23 February 2021). We chose to handle time in this way because during the lockdown periods, there were substantial changes to care and staffing: Staff were redeployed away from maternity settings, and hospital access restrictions changed [17]. Further details are provided in S1 Table.

The cohort included all women with a record in HES of a singleton birth during the pandemic period (23 March 2020 to 22 February 2021) and the pre-pandemic period (the corresponding calendar period 1 year earlier: 23 March 2019 to 22 February 2020). A birth (maternity) episode was defined as any record that contained information about mode of birth in procedure fields (OPCS-4 codes: R17.1 to R25.9), the main record, or the maternity tail. Multiple births were excluded. Multiple births were defined as birth episodes with an ICD-10 code for a multiple birth (Z37.2–7) or strong evidence of a multiple birth in the maternity tail (more than 1 distinct birthweight, birth order, and baby recorded in the same birth episode).

The study outcomes included stillbirth (fetal death at  $\geq$ 24 weeks' gestation), preterm birth (less than 37 weeks' gestation), small for gestational age (SGA) at birth (defined as birthweight < 10th centile using population-based centile charts [18]), induction of labour, mode of birth (instrumental vaginal birth, elective cesarean section, emergency cesarean section, or unassisted vaginal birth), prolonged maternal length of stay (3 or more days after

birth), and maternal readmission within 42 days of birth. For maternal readmissions, due to the 6-week follow-up required to assess the outcome, the cohort is restricted to births up to 17 February 2021 as the most recent HES dataset available at the time of analysis included records of hospital episodes until 31 March 2021.

Maternal age was grouped into 5-year periods, with women under 20 and over 39 years being aggregated into single categories. Parity and previous cesarean section were determined using women's previous HES hospital admission records. Ethnic background was coded using the Office for National Statistics (ONS) 2001 census categorisation for ethnicity, collapsed into 4 groups: White, South Asian, Black, and Other (combined Mixed and Other, including Chinese), as there is evidence of misclassification if more granular groups are used [19,20].

Information about comorbidities—including pre-existing and gestational diabetes mellitus, pre-existing hypertension, and preeclampsia—was available in the diagnosis codes in the HES records of the birth episode, with women assumed not to have these conditions if the relevant ICD-10 codes were not present. A woman was classified as having laboratory-confirmed SARS-CoV-2 infection at the time of birth if the ICD-10 code for 'COVID-19, virus identified' (U07.1) was recorded in the birth episode [21]. The test used to confirm infection in NHS hospital admissions is a nasal or throat swab examined using PCR [22].

The Index of Multiple Deprivation (IMD) was used as an overall measure of area-based deprivation. It is based on the income, education, employment, crime, and living environment in an individual's area of residence [23]. Quintiles of the national distribution of 2019 IMD rankings of 32,844 Lower Super Output Areas, each with typically 1,500 inhabitants, were used to categorise women into 5 groups according to socioeconomic background. Definitions of maternal characteristics and study outcomes, and their coding and completeness in HES, are outlined in S2 Table.

#### Statistical methods

All outcomes and analyses described in the Methods were prespecified, with the exception of the use of multiple imputation methods for missing data and case-mix adjustment. These modifications were done following editorial and peer review. We did not publish or pre-register an analysis plan. This study is reported as per the REporting of studies Conducted using Observational Routinely-collected Data (RECORD) guideline (S1 Text).

The characteristics of the women in the cohort were described for each analysis period, and monthly rates of different maternal and neonatal outcomes for the entire pre-pandemic and pandemic periods were plotted. Multi-level logistic regression models were used to estimate the differences in obstetric intervention and pregnancy outcome rates between the pandemic and the corresponding pre-pandemic calendar period in terms of odds ratios (ORs) and their 95% confidence intervals (CIs). ORs larger than 1 indicate that rates are increased during the pandemic period compared to the corresponding calendar period 1 year earlier. All analyses were adjusted for maternal age, obstetric history, comorbidities (pre-existing and gestational diabetes mellitus, pre-existing hypertension, and preeclampsia), and COVID-19 status at birth, and allowed for hospital-level clustering.

A binary variable for ethnicity (white versus all other categories, hereafter referred to as 'ethnic minority backgrounds') or socioeconomic background ('less deprived', defined as IMD quintiles 1 to 3, versus 'more deprived', defined as IMD quintiles 4 and 5) was included, and tests for interaction were used to determine if differences in outcome rates between pandemic and corresponding pre-pandemic periods varied according to ethnic and socioeconomic background. Missing values for maternal characteristics and outcomes were imputed using chained equations to generate 10 datasets; estimates from these datasets were pooled using Rubin's

rules [24]. A *p*-value of less than 0.05 was assumed to represent statistical significance. Stata 16 (StataCorp, College Station, TX) was used for all analyses.

## **Ethical approval**

This study used routinely collected administrative hospital data that were accessed to evaluate service provision and performance, and was therefore exempt from ethical review by the NHS Health Research Authority. The use of personal data without individual consent was approved by the NHS Health Research Authority (16/CAG/0058).

# Patient and public involvement

Neither women giving birth nor their representative organisations were explicitly consulted regarding this study. However, the outcome measures used were developed in collaboration with the National Maternity and Perinatal Audit (NMPA) Women and Families Group, and we will act in conjunction with this group to disseminate the findings of this study.

# Results

We identified 961,506 births in the English NHS during the pandemic period (23 March 2020 to 22 February 2021) and pre-pandemic period (23 March 2019 to 22 February 2020); 948,020 (98.6%) were singleton births. There was an overall 8.4% reduction in singleton births, from 493,293 in the pre-pandemic period to 451,727 in the pandemic period (S1 Fig). The characteristics of the women included were broadly similar in the pre-pandemic and pandemic periods (Table 1).

The monthly stillbirth rates are presented in Fig 1. Compared with the corresponding prepandemic calendar periods (Table 2), there were no statistically significant differences in stillbirth rates during the entire pandemic period (0.37% versus 0.36% pre-pandemic, adjusted OR [aOR] 0.95, 95% CI 0.89 to 1.02, p = 0.16), the first lockdown period (0.38% versus 0.35% in the same pre-pandemic calendar period, aOR 1.08, 95% CI 0.95 to 1.23, p = 0.24), or the second lockdown period (0.37% versus 0.38% in the same pre-pandemic calendar period, aOR 0.93, 95% CI 0.84 to 1.03, p = 0.18).

Compared with corresponding pre-pandemic figures (Table 2), there was a small reduction in preterm birth rates during the entire pandemic period (6.0% versus 6.1% pre-pandemic, aOR 0.96, 95% CI 0.94 to 0.97, p < 0.001), the first lockdown period (5.9% versus 6.1% pre-pandemic, aOR 0.95, 95% CI 0.92 to 0.99, p = 0.01), and the second lockdown period (6.1% versus 6.1% pre-pandemic, aOR 0.97, 95% CI 0.94 to 0.99, p = 0.01).

Compared with the corresponding pre-pandemic periods, SGA rates were lower for the entire pandemic period (5.6% versus 5.8% pre-pandemic, aOR 0.95, 95% CI 0.93 to 0.96, p < 0.001) and for the second lockdown period (Fig 1; Table 2; 5.3% versus 5.9% pre-pandemic, aOR 0.90, 95% CI 0.87 to 0.92, p < 0.001), but there was no statistically significant difference during the first lockdown period (5.8% versus 5.8% pre-pandemic, aOR 1.01, 95% CI 0.98 to 1.04, p = 0.58).

Induction of labour rates were higher during the pandemic period overall (40.4% versus 39.1% pre-pandemic, aOR 1.04, 95% CI 1.03 to 1.05, p < 0.001) and in the second lockdown period (40.8% versus 39.2%, aOR 1.05, 95% CI 1.03 to 1.06, p < 0.001), but not in the first lockdown period (39.4% in both periods, aOR 0.99, 95% CI 0.97 to 1.01, p = 0.24) (Fig 1; Table 2). Rates of elective and emergency cesarean section were higher overall in the pandemic period (aOR 1.13, 95% CI 1.11 to 1.14, and aOR 1.07, 95% CI 1.06 to 1.08, respectively; both p < 0.001), primarily driven by the higher rates in the second lockdown period (aOR 1.19, 95% CI 1.17 to 1.21, and aOR 1.11, 95% CI 1.09 to 1.12, respectively; both p < 0.001). Instrumental

Characteristic	Pre-pandemic	Pandemic	Pandemic periods				
			COVID-19 1st lockdown	COVID-19 local restrictions	COVID-19 2nd lockdown		
Time period covered	23 Mar 2019–22 Feb 2020	23 Mar 2020–22 Feb 2021	23 Mar 2020–23 Jun 2020	24 Jun 2020–21 Sep 2020	22 Sep 2020–22 Feb 2021		
Number of births	496,293	451,727	126,093	124,171	201,463		
Age (years)*							
≤19	14,361 (2.9)	12,005 (2.7)	3,431 (2.7)	3,418 (2.8)	5,156 (2.6)		
20-24	69,020 (13.9)	59,756 (13.2)	16,781 (13.3)	16,639 (13.4)	26,336 (13.1)		
25-29	138,118 (27.8)	123,572 (27.4)	34,644 (27.5)	34,362 (27.7)	54,566 (27.1)		
30-34	162,839 (32.8)	152,578 (33.8)	42,117 (33.4)	41,664 (33.6)	68,797 (34.1)		
35-39	91,400 (18.4)	84,141 (18.6)	23,674 (18.8)	22,669 (18.3)	37,798 (18.8)		
40+	20,550 (4.1)	19,671 (4.4)	5,444 (4.3)	5,419 (4.4)	8,808 (4.4)		
Obstetric history							
Primiparous	202,387 (40.8)	192,308 (42.6)	51,927 (41.2)	51,983 (41.9)	88,398 (43.9)		
Multiparous with no previous CS	233,736 (47.1)	207,275 (45.9)	58,525 (46.4)	57,157 (46.0)	91,593 (45.5)		
Multiparous with previous CS	60,170 (12.1)	52,144 (11.5)	15,641 (12.4)	15,031 (12.1)	21,472 (10.7)		
Comorbidities							
Pre-existing diabetes	4,612 (0.9)	4,211 (0.9)	1,161 (0.9)	1,143 (0.9)	1,907 (0.9)		
Gestational diabetes	40,177 (8.1)	39,215 (8.7)	10,304 (8.2)	10,649 (8.6)	18,262 (9.1)		
Pre-existing hypertension	3,497 (0.7)	3,384 (0.8)	889 (0.7)	957 (0.8)	1,538 (0.8)		
Preeclampsia or eclampsia	11,285 (2.3)	11,396 (2.5)	3,071 (2.4)	3,040 (2.5)	5,285 (2.6)		
COVID-19-positive at birth	0 (0)	4,578 (1.0)	564 (0.4)	199 (0.2)	3,815 (1.9)		
Ethnicity*							
White	336,484 (76.7)	307,799 (76.4)	85,096 (76.2)	83,934 (76.3)	138,769 (76.6)		
South Asian	52,056 (11.9)	49,423 (12.3)	13,798 (12.4)	13,448 (12.2)	22,177 (12.2)		
Black	20,517 (4.7)	18,344 (4.6)	5,190 (4.6)	5,085 (4.6)	8,059 (4.4)		
Other	29,365 (6.7)	27,459 (6.8)	7,607 (6.8)	7,608 (6.9)	12,244 (6.8)		
Socioeconomic deprivation*							
Quintile 1 = least deprived	73,691 (15.0)	68,480 (15.2)	19,003 (15.2)	18,495 (15.5)	30,982 (15.5)		
Quintile 2	83,614 (17.0)	77,359 (17.2)	21,498 (17.1)	21,045 (17.0)	34,816 (17.4)		
Quintile 3	94,921 (19.3)	86,637 (19.3)	24,124 (19.2)	23,820 (19.3)	38,693 (19.3)		
Quintile 4	109,876 (22.3)	99,831 (22.2)	27,863 (22.2)	27,476 (22.3)	44,492 (22.2)		
Quintile 5 = most deprived	130,446 (26.5)	116,874 (26)	32,900 (26.2)	32,614 (26.4)	51,270 (25.6)		

#### Table 1. Characteristics of pregnant women included in the study.

Data are given as n (percent) unless otherwise indicated. CS, cesarean section.

\*In the combined pre-pandemic and pandemic cohort, 9 (0.001%) records were missing information on maternal age, 106,583 (11.6%) records were missing information on ethnicity, and 6,381 (0.7%) records were missing information on Index of Multiple Deprivation.

https://doi.org/10.1371/journal.pmed.1003884.t001

births during the pandemic periods remained at similar levels to the corresponding pre-pandemic periods (Fig 1; Table 2).

The largest differences were seen for prolonged maternal length of stay (15.3% in the first lockdown versus 19.8% pre-pandemic, aOR 0.70, 95% CI 0.68 to 0.71, p < 0.001; 17.6% in the second lockdown versus 20.4% pre-pandemic, aOR 0.83, 95% CI 0.82 to 0.85, p < 0.001) and maternal readmission (2.6% in the first lockdown versus 3.3% pre-pandemic, aOR 0.77, 95% CI 0.73 to 0.81, p < 0.001; 3.1% in the second lockdown versus 3.3% pre-pandemic, aOR 0.91, 95% CI 0.88 to 0.95, p < 0.001) during the first and second lockdown periods compared to the corresponding pre-pandemic calendar periods. During the entire pandemic period, the rates



**Fig 1. Study outcome rates by month in the pre-pandemic (April 2019–February 2020) and pandemic (April 2020–February 2021) periods.** (A) Stillbirth; (B) preterm birth; (C) small for gestational age (SGA); (D) induction of labour; (E) elective cesarean section (CS); (F) emergency CS; (G) instrumental birth; (H) unassisted birth; (I) prolonged maternal le ngth of stay; (J) maternal readmission. lci, 95% lower confidence interval; uci, 95% upper confidence interval.

https://doi.org/10.1371/journal.pmed.1003884.g001

of prolonged maternal length of stay and readmission were also significantly lower than prepandemic rates (Fig 1; Table 2).

There was evidence that the differences in rates varied according to women's ethnic background for preterm births, cesarean births, and unassisted births (S3 Table), although this variation according to ethnic background was always small. The preterm birth rate was lower for white women during the pandemic period (6.1% pre-pandemic versus 5.9% during pandemic), but increased slightly for women of ethnic minority backgrounds (6.2% pre-pandemic to 6.3% during pandemic, *p*-value for interaction = 0.03). There was a greater increase in the rate of emergency cesarean birth for women of ethnic minority backgrounds (19.7% pre-pandemic to 21.9% during pandemic) compared to white women (15.9% pre-pandemic to 17.1% during pandemic, *p*-value for interaction = 0.02). There was a greater reduction in unassisted births for women of ethnic minority backgrounds than for white women (*p*-value for interaction < 0.001). Otherwise, the differences were similar in women from white and ethnic minority backgrounds.

There was no evidence that the differences in intervention rates and pregnancy outcomes varied according to socioeconomic background (all *p*-values for interaction > 0.05; <u>S4 Table</u>).

## Discussion

In this national study, we found that, overall, pregnancy outcomes during the COVID-19 pandemic were similar to outcomes in the corresponding calendar period 1 year earlier. There were small decreases in preterm birth and SGA birth rates and small increases in induction of labour and elective and emergency cesarean section rates during the pandemic period. Furthermore, there were lower rates of prolonged maternal length of stay and 42-day readmission, especially during the first lockdown period. Lastly, there was some evidence of a slightly different pattern of results in women from ethnic minority backgrounds, with a small increase in

Outcome	Measure	Pandemic versus pre-pandemic period(time period covered: 23 March-22 February)			COVID-19 first lockdown versus previous year (time period covered: 23 March-23 June)			COVID-19 second lockdown versus previous year (time period covered: 22 September–22 February)					
		Pre- pandemic	Pandemic	OR (95% CI) <i>p</i> -value	aOR* (95% CI) <i>p</i> -value	Pre- pandemic	Pandemic	OR (95% CI) <i>p</i> -value	aOR* (95% CI) <i>p</i> -value	Pre- pandemic	Pandemic	OR (95% CI) p-value	aOR* (95% CI) <i>p</i> -value
Number of observations		493,293	451,727			136,161	126,093			224,014	201,463		
Stillbirth	Num/ denom	1,818/ 496,293	1,605/ 451,727	0.97 (0.91, 1.04)	0.95 (0.89, 1.02) p = 0.16	477/136,161	481/126,093	$ \begin{array}{cccc} 1.09 & (0.96, & 1 \\ 1.24) & 1 \\ p = 0.18 & p \end{array} $	1.08 (0.95, 1.23) p = 0.24	845/224,014	736/201,463	$\begin{array}{c} 0.97 \ (0.88, \\ 1.07) \\ p = 0.58 \end{array}$	0.93 (0.84, 1.03)
	Percent	0.37	0.36	p = 0.44		0.35	0.38			0.38	0.37		<i>p</i> = 0.18
Preterm birth	Num/ denom	29,983/ 492,076	25,633/ 429,408	$\begin{array}{c} 0.98 \ (0.96, \\ 1.00) \\ p = 0.02 \end{array}$	$0.96 (0.94, \\ 0.97) \\ p < 0.001$	8,246/ 134,782	7,403/ 124,907	0.97 (0.94, 1.00) p = 0.05	0.95 (0.92, 0.99) p = 0.01	13,649/ 22,214	11,082/ 181,609	0.99 (0.97, 1.02) p = 0.62	0.97 (0.94, 0.99)
	Percent	6.1	6.0			6.1	5.9			6.1	6.1		<i>p</i> = 0.01
Small for gestational age	Num/ denom	28,514/ 489,054	23,681/ 426,497	$\begin{array}{c} 0.95 \ (0.94, \\ 0.97) \\ p < 0.001 \end{array}$	$\begin{array}{c} 0.95 \ (0.93, \\ 0.96) \\ p < 0.001 \end{array}$	7,701/ 133,949	7,210/ 123,966	$ \begin{array}{c cccc} 1.01 & (0.98, & 1.0 \\ 1.08) & 1.0 \\ p = 0.45 & p = \end{array} $	1.01 (0.98, 1.04)	13,007/ 220,811	9,552/ 180,496	$\begin{array}{c} 0.91 \ (0.88, \\ 0.93) \\ p < 0.001 \end{array}$	0.90 (0.87, 0.92) <i>p</i> < 0.001
	Percent	5.8	5.6			5.8	5.8		p = 0.58	5.9	5.3		
Induction of labour**	Num/ denom	135,937/ 348,007	125,176/ 310,135	$ \begin{array}{c} 1.04 (1.03, \\ 1.05) \\ p < 0.001 \end{array} $	$ \begin{array}{c} 1.04 (1.03, \\ 1.05) \\ p < 0.001 \end{array} $	37,468/ 95,180	34,348/ 87,225	$\begin{array}{ccc} 1.00 \ (0.98, & 0.9 \\ 1.02) & 1.0 \\ p = 0.82 & p = 0.82 \end{array}$	0.99 (0.97, 1.01)	61,461/ 156,842	55,800/ 136,880	$ \begin{array}{c} 1.05 (1.04, \\ 1.07) \\ p < 0.001 \end{array} $	1.05 (1.03, 1.06) <i>p</i> < 0.001
	Percent	39.1	40.4			39.4	39.4		p = 0.24	39.2	40.8		
Elective cesarean section	Num/ denom	64,079/ 496,293	62,552/ 451,727	$ \begin{array}{c c} 1.08 (1.07, \\ 1.09) \\ p < 0.001 \end{array} $	$ \begin{array}{c} 1.13 (1.11, \\ 1.14) \\ p < 0.001 \end{array} $	17,020/ 136,161	17,045/ 126,093	$\begin{array}{c cccc} 1.09 & (1.07, & 1 \\ 1.12) & 1 \\ p < 0.001 & p \end{array}$	1.08 (1.06, 1.11)	29,529/ 224,014	28,539/ 201,463	$ \begin{array}{c} 1.08 (1.07, \\ 1.10) \\ p < 0.001 \end{array} $	1.19 (1.17, 1.21) <i>p</i> < 0.001
	Percent	12.9	13.9			12.5	13.5		p < 0.001	13.2	14.2		
Emergency cesarean section	Num/ denom	84,156/ 496,293	82,975/ 451,727	$\begin{array}{c} 1.10 \ (1.09, \\ 1.11) \\ p < 0.001 \end{array}$	1.07 (1.06, 1.08) <i>p</i> < 0.001	22,974/ 136,161	21,841/ 126,093	1.03 (1.01, 1.05) p = 0.01	1.00 (0.98, 1.02) p = 0.95	38,595/ 224,014	38,623/ 201,463	$ \begin{array}{c} 1.14 (1.12, \\ 1.16) \\ p < 0.001 \end{array} $	1.11 (1.09, 1.12) <i>p</i> < 0.001
	Percent	17.0	18.4			16.9	17.3			17.2	19.2		
Instrumental birth	Num/ denom	61,256/ 496,293	57,602/ 451,727	$ \begin{array}{c} 1.04 (1.02, \\ 1.05) \\ p < 0.001 \end{array} $	1.01 (1.00, 1.02)  p = 0.16	16,427/ 136,161	15,975/ 126,093	$ \begin{array}{c} 1.06 (1.03, \\ 1.08) \\ p < 0.001 \end{array} $	1.04 (1.01, 1.06) <i>p</i> = 0.002	27,807/ 224,014	25,733/ 201,463	$ \begin{array}{c} 1.03 (1.01, \\ 1.05) \\ p = 0.002 \end{array} $	0.99 (0.97, 1.01) p = 0.40
	Percent	12.3	12.8			12.1	12.7			12.4	12.9		
Unassisted birth	Num/ denom	285,525/ 496,293	247,474/ 451,727	0.90 (0.89, 0.90) <i>p</i> < 0.001	$\begin{array}{c cccc} 0.90 & (0.89, & 79,4 \\ 0.91) & 136, \\ p < 0.001 & 58.3 \end{array}$	79,416/ 136,161	70,914/ 126,093	$ \begin{array}{c c} 0.92 \ (0.91, & 0 \\ 0.93) & 0 \\ p < 0.001 & p \end{array} $	0.95 (0.93, 0.96)	127,517/ 224,014	108,091/ 201,463	$\begin{array}{c} 0.88 \ (0.87, \\ 0.89) \\ p < 0.001 \end{array}$	0.87 (0.86, 0.88) p < 0.001
	Percent	57.5	54.8			58.3	56.2		<i>p</i> < 0.001	56.9	53.7		
Prolonged maternal length of stay	Num/ denom	96,177/ 481,458	73,047/ 437,573	$\begin{array}{c cccc} 0.80 & (0.79, & 0 \\ 0.81) & 0 \\ p < 0.001 & p \end{array}$	0.77 (0.76, 0.78) <i>p</i> < 0.001	26,199/ 132,143	18,742/ 122,872	$\begin{array}{c} 0.73 \ (0.71, \\ 0.74) \\ p < 0.001 \end{array}$	0.70 (0.68, 0.71) <i>p</i> < 0.001	44,299/ 217,319	34,165/ 194,298	$\begin{array}{c c} 0.83 \ (0.82, \\ 0.85) \\ p < 0.001 \end{array}$	0.80 (0.78, 0.81)
	Percent	20.0	16.7			19.8	15.3			20.4	17.6		<i>p</i> < 0.001
Maternal readmission**	Num/ denom	15,888/ 418,393	12,725/ 431,164	$ \begin{array}{c} 0.89 \ (0.87, \\ 0.91) \\ p < 0.001 \end{array} $	$\begin{array}{c} 0.88 \ (0.86, \\ 0.90) \\ p < 0.001 \end{array}$	4,387/ 132,120	3,161/ 122,851	$\begin{array}{c c} 0.77 \ (0.73, \\ 0.81) \\ p < 0.001 \end{array}$	0.75 (0.72, 0.79)	7,172/ 217,289	5,802/ 187,920	$ \begin{array}{c c} 0.93 \ (0.90, \\ 0.97) \\ p < 0.001 \end{array} \begin{array}{c} 0 \\ p \end{array} $	0.91 (0.88, 0.95)
	Percent	3.3	3.0			3.3	2.6		<i>p</i> < 0.001	3.3	3.1		<i>p</i> < 0.001

Table 2.	Comparisons of outcomes in th	COVID-19 pandemic and lockdown	n periods with the same	periods in the previous	year.

aOR, adjusted odds ratio; CI, confidence interval; Num/denom, numerator/denominator.

\*aORs from multi-level logistic regression models adjusted for maternal age, obstetric history, and comorbidities. All missing values for maternal characteristics and outcomes were imputed. *p*-Value from *t* test for the null hypothesis that the OR is equal to 1.

\*\*Induction of labour denominator is restricted to women who did not have an elective cesarean section; maternal readmission denominator is restricted to women who gave birth up to and including 17 February 2021 and were discharged within 42 days of delivery.

https://doi.org/10.1371/journal.pmed.1003884.t002

existing differences in the rates of preterm birth, emergency cesarean section, and unassisted vaginal delivery.

Our study concurs with preliminary reports from the ONS that rates of stillbirth in England were unchanged during the COVID-19 pandemic [10]. Moreover, we observed a lower number of singleton births overall during the pandemic compared to the pre-pandemic period, a phenomenon also noted by the ONS. However, unlike the ONS, which uses national birth registration data, our study found a statistically significant reduction in the rates of preterm birth and babies born SGA.

These observed differences between population-level analyses of the same population may partly be attributable to our analysis being restricted to singleton births. While multiple births represent a minority of all births (1.4%) [25], over half of multiple pregnancies result in birth before 37 weeks' gestation [26]. In addition, it cannot be excluded that those undergoing 'selective fertility' (e.g., women requiring assisted conception), were less likely to become pregnant

during these early months of the pandemic periods, giving birth in the later pandemic periods that were not covered by earlier studies in the UK [27].

This study considers interventions and outcomes at the time of birth that may have been affected by circumstances at different points throughout pregnancy. For example, fetal growth (measured as the proportion of babies born SGA) will reflect not only the impact of COVID-19 restrictions and clinical care experienced during the entire antenatal period, but also guidance by national maternity initiatives to reduce preterm birth and stillbirth [28–31]. In addition, there is a pre-existing national trend towards an increase in induction of labour, and this change in practice may have affected the results that we report [32]. Unfortunately, the dataset used does not contain information about antenatal care, and therefore it is not possible to directly attribute changes in outcomes to changes that have occurred as a result of the pandemic.

The reduction in preterm birth rate during the pandemic period was also demonstrated in an earlier UK-based study reporting neonatal admission rates [33]. Possible mechanisms for the observed reduction in preterm birth include modified population behaviour, resulting in reduced exposure to other pathogens including influenza, lower levels of physical exercise, and reduced workplace stress [34,35]. Similar findings have been observed in the preterm birth rate in China, with the rate returning to pre-pandemic levels after pandemic restrictions on societal interactions were lifted [36,37]. Changes in clinical practice due to the pandemic, such as reductions in labour induction and cesarean birth at earlier gestational age—perhaps due to a reduction in the identification of fetal growth restriction or preeclampsia due to reduced in-person appointments—could have also contributed to lower rates of preterm births [33].

During the pandemic period, our study population experienced higher rates of interventions including induction of labour and elective and emergency cesarean birth. To the best of our knowledge, similar findings have not been reported by other studies [38–40]. Although it is difficult to unpick the complex relationship between all factors influencing maternity care during the COVID-19 pandemic, it is likely that our findings of increased interventions are partially attributable to the centralisation of maternity services, including closure of midwifeled birth settings, which are associated with reduced interventions for women at low risk of complications [13,41,42]. Other potential explanations are that changes in care-seeking behaviour amongst pregnant women and an increase in virtual appointments may have created delays in the diagnosis of adverse conditions, which may have led in turn to missed opportunities for conservative management. Also, there may also have been a change in clinician behaviour towards a lower threshold for interventions in order to expedite births and to avoid emergency situations. Lastly, occupational stress and burnout among maternity staff could also be associated with a more interventionist approach [43].

Our results show a reduction in prolonged maternal hospital stays, a finding supported by other studies [44–46]; early discharge from the hospital may have been initiated by either the clinician or the woman to reduce the perceived risk of infection associated with hospital stay, or to reunite with family members unable to visit maternity units. Rates of maternal readmission were also reduced during the pandemic period, which may reflect that women were less likely to seek care for postpartum issues.

There is evidence from other clinical areas that people with ethnic minority backgrounds and those from more deprived communities were disproportionately affected by the COVID-19 pandemic [6,47,48]. Our results demonstrate little variation according to women's ethnic background in the differences associated with the pandemic periods. This may be explained by recommendations from UK midwifery and obstetric professional bodies to be particularly attentive to women in these groups, given that they are considered to be at higher risk of complications from COVID-19 [12].

The main strengths of this study are its large size, with the study including almost all births in England over a period of almost 2 years, covering periods of high COVID-19 prevalence in the UK and capturing significant events along the pandemic timeline. HES data are well-established and have previously been used for national comparisons of maternity care [49]. The clinical and patient demographic information in HES (for both maternity and further inpatient episodes after birth) allowed for a broader range of maternity care and neonatal outcomes to be compared across COVID-19 periods than previous studies exploring the combined direct and indirect effects of the pandemic [7,39]. Access to individual data for each woman and infant, rather than national aggregate data, enabled comparison of differences in intervention and outcome rates according to women's ethnic and socioeconomic backgrounds.

This study reports the differences in the rates of a number of interventions and outcomes between pandemic and pre-pandemic periods, also exploring whether these differences varied according to ethnic and socioeconomic background. These multiple comparisons, increasing the chance of false-positive results, should be taken into account, especially when interpreting small but statistically significant differences [50].

Our study raises questions about the potential associations between societal and care changes during the COVID-19 pandemic and maternal and perinatal outcomes. The pandemic can be regarded as a natural experiment in which many aspects of care and behaviour changed simultaneously. Our results suggest that these changes may have led to reductions in the rate of preterm and SGA birth as well as increases in obstetric intervention rates. All this requires further investigation, particularly as similar results have not, to the best of our knowledge, been seen in other high-income settings [7].

#### Conclusion

In this study of all births in the English NHS, we found a small decrease in the rates of preterm and SGA birth during the pandemic, compared to the corresponding pre-pandemic calendar period 1 year earlier. There was a small increase in the rates of obstetric interventions, which may be a consequence of temporary closure of midwife-led birth settings or a response to delays in pregnant women seeking care. These results varied slightly according to women's ethnic background, but not according to their socioeconomic background. All these findings should be interpreted with caution, because differences were small, many comparisons were made, and the relationships between the factors influencing maternity care and pregnancy outcomes during the COVID-19 pandemic are complex.

# Supporting information

**S1 Fig. Number of singleton births, per month.** (TIF)

S1 Table. Timeline of COVID-19 restrictions across England and the United Kingdom from 23 March 2020 to 22 February 2021. (DOC)

**S2** Table. Definitions of maternal characteristics and maternal and perinatal outcomes, and their coding and completeness in Hospital Episode Statistics (HES). Table A: Maternal characteristics. Table B: Maternal and perinatal outcomes. (DOC)

S3 Table. Comparisons of outcomes in the COVID-19 pandemic with outcomes in the same period in the previous year, by ethnic background. (DOC)

S4 Table. Comparisons of outcomes in the COVID-19 pandemic with outcomes in the same period in the previous year, by deprivation. (DOC)

S1 Text. RECORD statement—Checklist of items, extended from the STROBE statement, that should be reported in observational studies using routinely collected health data. (DOC)

# **Author Contributions**

Conceptualization: Ipek Gurol-Urganci, Lara Waite, Kirstin Webster, Jennifer Jardine.

Data curation: Ipek Gurol-Urganci, Jennifer Jardine.

Formal analysis: Ipek Gurol-Urganci, Jennifer Jardine.

Funding acquisition: Jan van der Meulen.

Methodology: Jan van der Meulen.

Writing - original draft: Ipek Gurol-Urganci, Lara Waite, Kirstin Webster, Jennifer Jardine.

Writing – review & editing: Ipek Gurol-Urganci, Lara Waite, Kirstin Webster, Jennifer Jardine, Fran Carroll, George Dunn, Alissa Frémeaux, Tina Harris, Jane Hawdon, Patrick Muller, Jan van der Meulen, Asma Khalil.

#### References

- Gurol-Urganci I, Jardine JE, Carroll F, Draycott T, Dunn G, Fremeaux A, et al. Maternal and perinatal outcomes of pregnant women with SARS-CoV-2 infection at the time of birth in England: national cohort study. Am J Obstet Gynecol. 2021; 225:e1–11. https://doi.org/10.1016/j.ajog.2021.05.016 PMID: 34023315
- MBRRACE-UK. Key information on COVID-19 in pregnancy. Oxford: University of Oxford National Perinatal Epidemiology Unit; 2021 [cited 2021 Oct 13]. https://www.npeu.ox.ac.uk/assets/downloads/ MBRRACE-UK\_Rapid\_COVID\_19\_2021 -\_\_Infographic\_v10.pdf.
- Papageorghiou AT, Deruelle P, Gunier RB, Rauch S, García-May PK, Mhatre M, et al. Preeclampsia and COVID-19: results from the INTERCOVID prospective longitudinal study. Am J Obstet Gynecol. 2021; 225:289.e1–289.e17. https://doi.org/10.1016/j.ajog.2021.05.014 PMID: 34187688
- 4. Villar J, Ariff S, Gunier RB, Thiruvengadam R, Rauch S, Kholin A, et al. Maternal and neonatal morbidity and mortality among pregnant women with and without COVID-19 infection. JAMA Pediatr. 2021; 175:817. https://doi.org/10.1001/jamapediatrics.2021.1050 PMID: 33885740
- Vousden N, Bunch K, Morris E, Simpson N, Gale C, O'Brien P, et al. The incidence, characteristics and outcomes of pregnant women hospitalized with symptomatic and asymptomatic SARS-CoV-2 infection in the UK from March to September 2020: a national cohort study using the UK Obstetric Surveillance System (UKOSS). PLoS ONE. 2021; 16:e0251123. https://doi.org/10.1371/journal.pone.0251123 PMID: 33951100
- Mathur R, Rentsch CT, Morton CE, Hulme WJ, Schultze A, MacKenna B, et al. Ethnic differences in SARS-CoV-2 infection and COVID-19-related hospitalisation, intensive care unit admission, and death in 17 million adults in England: an observational cohort study using the OpenSAFELY platform. Lancet. 2021; 397:1711–24. https://doi.org/10.1016/S0140-6736(21)00634-6 PMID: 33939953
- Chmielewska B, Barrett I, Townsend R, Kalafat E, van der Meulen J, Gurol-Urganci I, et al. Impact of the COVID-19 pandemic on maternal and perinatal outcomes: a systematic review and meta-analysis. Lancet Glob Health. 2021; 9:e759–72. <u>https://doi.org/10.1016/S2214-109X(21)00079-6</u> PMID: 33811827
- Roberton T, Carter ED, Chou VB, Stegmuller AR, Jackson BD, Tam Y, et al. Early estimates of the indirect effects of the COVID-19 pandemic on maternal and child mortality in low-income and middle-income countries: a modelling study. Lancet Glob Health. 2020; 8:901–9. https://doi.org/10.1016/ S2214-109X(20)30229-1 PMID: 32405459

- Renfrew MJ, Cheyne H, Craig J, Duff E, Dykes F, Hunter B, et al. Sustaining quality midwifery care in a pandemic and beyond. Midwifery. 2020; 88:102759. <u>https://doi.org/10.1016/j.midw.2020.102759</u>
   PMID: 32485502
- Office for National Statistics. Provisional births in England and Wales: 2020. Newport: Office for National Statistics; 2020 [cited 2021 Mar 16]. https://www.ons.gov.uk/peoplepopulationandcommunity/ birthsdeathsandmarriages/livebirths/articles/provisionalbirthsinenglandandwales/2020.
- Institute for Government. Timeline of UK coronavirus lockdowns, March 2020 to March 2021. London: Institute for Government; 2021 [cited 2021 Mar 12]. https://www.instituteforgovernment.org.uk/sites/ default/files/timeline-lockdown-web.pdf.
- Royal College of Obstetricians & Gynaecologists, Royal College of Midwives. Coronavirus (COVID-19) infection in pregnancy. Information for healthcare professionals. Version 13. London: Royal College of Obstetricians & Gynaecologists; 2021 Feb 19.
- Coxon K, Turienzo CF, Kweekel L, Brigante L, Simon A, Lanau MM. The impact of the coronavirus (COVID-19) pandemic on maternity care in Europe. Midwifery. 2020; 88:102779. <u>https://doi.org/10.1016/j.midw.2020.102779 PMID: 32600862</u>
- NHS Digital. Hospital Episode Statistics (HES). Leeds: NHS Digital; 2020 [cited 2021 Mar 12]. https:// digital.nhs.uk/data-and-information/data-tools-and-services/data-services/hospital-episode-statistics.
- World Health Organization. ICD-10 version: 2010. Geneva: World Health Organization; 2010 [cited 2021 Mar 12]. https://icd.who.int/browse10/2010/en.
- National Health Service. OPCS classification of interventions and procedures. London: National Health Service; 2020 [cited 2021 Mar 12]. https://datadictionary.nhs.uk/supporting\_information/opcs\_ classification of interventions and procedures.html.
- Priddy S. The coronavirus timeline: measures taken by the House of Commons. London: UK Parliament; 2021 [cited 2021 Oct 9]. <u>https://commonslibrary.parliament.uk/house-of-commons-coronavirustimeline/.</u>
- Cole TJ. The development of growth references and growth charts. Ann Hum Biol. 2012; 39:382–94. https://doi.org/10.3109/03014460.2012.694475 PMID: 22780429
- Gov.UK. List of ethnic groups. Gov.UK; 2020 [cited 2020 Dec 7]. https://www.ethnicity-facts-figures. service.gov.uk/style-guide/ethnic-groups.
- Jardine JE, Frémeaux A, Coe M, Gurol Urganci I, Pasupathy D, Walker K. Validation of ethnicity in administrative hospital data in women giving birth in England: cohort study. BMJ Open. 2021; 11: e051977. https://doi.org/10.1136/bmjopen-2021-051977 PMID: 34426472
- World Health Organization. Emergency use ICD codes for COVID-19 disease outbreak. Geneva: World Health Organization; 2021 [cited 2021 Oct 12]. https://www.who.int/standards/classifications/ classification-of-diseases/emergency-use-icd-codes-for-covid-19-disease-outbreak.
- NHS England, NHS Improvement. Healthcare associated COVID-19 infections—further action. London: NHS England; 2020 [cited 2022 Jan 5]. https://www.england.nhs.uk/coronavirus/wp-content/ uploads/sites/52/2020/06/Healthcare-associated-COVID-19-infections—further-action-24-June-2020. pdf.
- 23. Ministry of Housing, Communities & Local Government. English indices of deprivation 2019: technical report. London: Ministry of Housing, Communities & Local Government; 2019 [cited 2022 Jan 5]. https://www.gov.uk/government/publications/english-indices-of-deprivation-2019-technical-report.
- White IR, Royston P, Wood AM. Multiple imputation using chained equations: issues and guidance for practice. Stat Med. 2011; 30:377–99. https://doi.org/10.1002/sim.4067 PMID: 21225900
- 25. Office for National Statistics. Birth characteristics in England and Wales: 2018. Newport: Office for National Statistics; 2019 [cited 2022 Jan 5]. https://www.ons.gov.uk/peoplepopulationandcommunity/ birthsdeathsandmarriages/livebirths/bulletins/birthcharacteristicsinenglandandwales/2018.
- 26. NMPA Project Team. NHS maternity care for women with multiple births and their babies: a study on feasibility of assessing care using data from births between 1 April 2015 and 31 March 2017 in England, Wales and Scotland. London: National Maternity and Perinatal Audit; 2020 [cited 2022 Jan 5]. https://maternityaudit.org.uk/FilesUploaded/NMPA%20Multiple%20Births%20Report%202020.pdf.
- Smith ADAC, Gromski PS, Al Rashid K, Tilling K, Lawlor DA, Nelson SM. Population implications of cessation of IVF during the COVID-19 pandemic. Reprod Biomed Online. 2020; 41:428–30. <u>https://doi.org/ 10.1016/j.rbmo.2020.07.002</u> PMID: 32753362
- NHS England. Maternity transformation programme. London: NHS England; 2021 [cited 2021 Oct 8]. https://www.england.nhs.uk/mat-transformation/.
- National Health Service. The NHS long term plan. London: National Health Service; 2019 [cited 2022 Jan 5]. https://www.longtermplan.nhs.uk/wp-content/uploads/2019/08/nhs-long-term-plan-version-1.2. pdf.

- NHS England. Saving babies' lives version two: a care bundle for reducing perinatal mortality. London: NHS England; 2019 [cited 2022 Jan 5]. https://www.england.nhs.uk/wp-content/uploads/2019/03/ Saving-Babies-Lives-Care-Bundle-Version-Two-Updated-Final-Version.pdf.
- **31.** NHS Resolution. Maternity incentive scheme—year four. London: NHS Resolution; 2021 [cited 2022 Jan 5]. https://resolution.nhs.uk/wp-content/uploads/2021/08/MIS-Y4-guidance.pdf.
- Selvaratnam R, Davey M, Mol B, Wallace E. Increasing obstetric intervention for fetal growth restriction is shifting birthweight centiles: a retrospective cohort study. BJOG. 2020; 127:1074–80. https://doi.org/ 10.1111/1471-0528.16215 PMID: 32180311
- 33. Greenbury SF, Longford N, Ougham K, Angelini ED, Battersby C, Uthaya S, et al. Changes in neonatal admissions, care processes and outcomes in England and Wales during the COVID-19 pandemic: a whole population cohort study. BMJ Open. 2021; 11:e054410. https://doi.org/10.1136/bmjopen-2021-054410 PMID: 34598993
- Rasmussen SA, Jamieson DJ, Uyeki TM. Effects of influenza on pregnant women and infants. Am J Obstet Gynecol. 2012; 207(3 Suppl):S3–S8. https://doi.org/10.1016/j.ajog.2012.06.068 PMID: 22920056
- 35. Philip RK, Purtill H, Reidy E, Daly M, Imcha M, McGrath D, et al. Unprecedented reduction in births of very low birthweight (VLBW) and extremely low birthweight (ELBW) infants during the COVID-19 lock-down in Ireland: a 'natural experiment' allowing analysis of data from the prior two decades. BMJ Glob Health. 2020; 5:e003075. https://doi.org/10.1136/bmjgh-2020-003075 PMID: 32999054
- Bian Z, Qu X, Ying H, Liu X. Are COVID-19 mitigation measures reducing preterm birth rate in China? BMJ Glob Health. 2021; 6:e006359. https://doi.org/10.1136/bmjgh-2021-006359 PMID: 34385161
- 37. Gallo LA, Gallo TF, Borg DJ, Moritz KM, Clifton VL, Kumar S. A decline in planned, but not spontaneous, preterm birth rates in a large Australian tertiary maternity centre during COVID-19 mitigation measures. Aust N Z J Obstet Gynaecol. 2021 Jul 12. https://doi.org/10.1111/ajo.13406 PMID: 34254286
- Bhatia K, Columb M, Bewlay A, Eccles J, Hulgur M, Jayan N, et al. The effect of COVID-19 on general anaesthesia rates for caesarean section. A cross-sectional analysis of six hospitals in the north-west of England. Anaesthesia. 2021; 76:312–9. https://doi.org/10.1111/anae.15313 PMID: 33073371
- Khalil A, von Dadelszen P, Draycott T, Ugwumadu A, O'Brien P, Magee L. Change in the incidence of stillbirth and preterm delivery during the COVID-19 pandemic. JAMA. 2020; 324:705. <u>https://doi.org/10.1001/jama.2020.12746 PMID: 32648892</u>
- Gemmill A, Casey JA, Catalano R, Karasek D, Margerison CE, Bruckner T. Changes in preterm birth and caesarean deliveries in the United States during the SARS-CoV-2 pandemic. Paediatr Perinat Epidemiol. 2021 Sep 13. https://doi.org/10.1111/ppe.12811 PMID: 34515360
- 41. Birthplace in England Collaborative Group, Brocklehurst P, Hardy P, Hollowell J, Linsell L, Macfarlane A, et al. Perinatal and maternal outcomes by planned place of birth for healthy women with low risk pregnancies: the Birthplace in England national prospective cohort study. BMJ. 2011; 343:d7400. https://doi.org/10.1136/bmj.d7400 PMID: 22117057
- Jardine J, Relph S, Magee L, Dadelszen P, Morris E, Ross-Davie M, et al. Maternity services in the UK during the coronavirus disease 2019 pandemic: a national survey of modifications to standard care. BJOG. 2021; 128:880–89. https://doi.org/10.1111/1471-0528.16547 PMID: 32992408
- Bourne T, Shah H, Falconieri N, Timmerman D, Lees C, Wright A, et al. Burnout, well-being and defensive medical practice among obstetricians and gynaecologists in the UK: cross-sectional survey study. BMJ Open. 2019; 9:e030968. https://doi.org/10.1136/bmjopen-2019-030968 PMID: 31767585
- Bornstein E, Gulersen M, Husk G, Grunebaum A, Blitz MJ, Rafael TJ, et al. Early postpartum discharge during the COVID-19 pandemic. J Perinat Med. 2020; 48:1008–12. <u>https://doi.org/10.1515/jpm-2020-0337 PMID: 32845868</u>
- 45. Greene NH, Kilpatrick SJ, Wong MS, Ozimek JA, Naqvi M. Impact of labor and delivery unit policy modifications on maternal and neonatal outcomes during the coronavirus disease 2019 pandemic. Am J Obstet Gynecol MFM. 2020; 2:100234. https://doi.org/10.1016/j.ajogmf.2020.100234 PMID: 32984804
- 46. Kugelman N, Toledano-Hacohen M, Karmakar D, Segev Y, Shalabna E, Damti A, et al. Consequences of the COVID-19 pandemic on the postpartum course: lessons learnt from a large-scale comparative study in a teaching hospital. Int J Gynaecol Obstet. 2021; 153:315–21. <u>https://doi.org/10.1002/ijgo.13633 PMID: 33523481</u>
- 47. Knight M, Bunch K, Cairns A, Cantwell R, Cox P, Kenyon S, et al. Saving lives, improving mothers' care. Rapid report: learning from SARS-CoV-2-related and associated maternal deaths in the UK. Oxford: University of Oxford National Perinatal Epidemiology Unit; 2020 [cited 2022 Jan4]. https://www.npeu.ox.ac.uk/assets/downloads/mbrrace-uk/reports/MBRRACE-UK\_Maternal\_Report\_2020\_v10\_FINAL.pdf.
- 48. Thomson K, Moffat M, Arisa O, Jesurasa A, Richmond C, Odeniyi A, et al. Socioeconomic inequalities and adverse pregnancy outcomes in the UK and Republic of Ireland: a systematic review and meta-

analysis. BMJ Open. 2021; 11:e042753. https://doi.org/10.1136/bmjopen-2020-042753 PMID: 33722867

- 49. Royal College of Obstetricians & Gynaecologists. Patterns of maternity care in English NHS trusts: 2013/14. London: Royal College of Obstetricians & Gynaecologists; 2016 [cited 2022 Jan 5]. https:// www.rcog.org.uk/globalassets/documents/guidelines/research—audit/maternity-indicators-2013-14\_ report2.pdf.
- 50. Althouse AD. Adjust for multiple comparisons? It's not that simple. Ann Thorac Surg. 2016; 101:1644– 5. https://doi.org/10.1016/j.athoracsur.2015.11.024 PMID: 27106412